Author's response to reviews

Title: Current level and determinants of inappropriate admissions to township hospitals under the New Rural Cooperative Medical System in China: a cross-sectional study

Authors:

Zhang Yan Dr. (yanzhang@hust.edu.cn)
Yingchun Chen Pro. (chenyingchun@mail.hust.edu.cn)
Xiang Zhang Pro. (zhangchxiang@163.com)
Liang Zhang Pro. (zhangliang@ mails.tjmu.edu.cn)

Version: 6  Date: 20 November 2014

Author's response to reviews: see over
Response to editor’s comments

Dear editor,

Thanks for your incredible work for reviewing our manuscript these months. Firstly we would like to express our gratitude to you for your valuable input. Secondly we have made some revisions in accordance with your comments, and would like to re-submit the revised manuscript.

Our responses to your comments are below:

1. Please delete the funding section and add the funding information to the Acknowledgements section.
   A: We have deleted the funding section and adjusted the Acknowledgements section as you required.

2. Please do not upload cover letters as additional files because they form attachments to the main manuscript.
   A: We have uploaded this cover letters as Response to Reviewers.

3. Please also ensure that your revised manuscript conforms to the journal style.
   A: We have adjusted the revised manuscript conforms to the journal style in point-by-point response, including the font, format, reference, and tables.